

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Crowley, Meredith

## Working Paper Cyclical dumping and US antidumping protection: 1980 - 2001

Working Paper, No. 2007-21

**Provided in Cooperation with:** Federal Reserve Bank of Chicago

*Suggested Citation:* Crowley, Meredith (2009) : Cyclical dumping and US antidumping protection: 1980 - 2001, Working Paper, No. 2007-21, Federal Reserve Bank of Chicago, Chicago, IL

This Version is available at: https://hdl.handle.net/10419/70580

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

## Cyclical Dumping and US Antidumping Protection: 1980-2001

Meredith A. Crowley<sup>1</sup>

Economic Research Federal Reserve Bank of Chicago March 2009

#### Abstract

This paper estimates the impact of economic conditions in foreign industries on the filing of antidumping petitions by US industries and the US government's decision in preliminary and final antidumping investigations. Exploiting cross-country variation in economic shocks in manufacturing, I estimate a joint model of filing decisions by the US industry and antidumping decisions by the US government.

I find strong evidence that economic weakness in a foreign industry is associated with an increase in the probability of antidumping protection. After controlling for other political and economic factors that likely drive industry filing and government decisions including US GDP growth, I find that a one standard deviation fall in the growth of employment (consumption) in a foreign economy's manufacturing industry doubles (triples) the joint probability that the US industry will file an antidumping petition and the US government will impose a preliminary (temporary) antidumping measure. The effect of weakness in a foreign economy is even larger for final antidumping measures. A one standard deviation fall in foreign employment (consumption) growth increases the joint probability that a petition will be filed and a final (long-lasting) antidumping measure will be imposed by a factor of five (seven). In finding that US trade policy is applied counter-cyclically to foreign economic fluctuations, the paper suggests that trade policy may reduce the extent of business cycle transmission across countries.

JEL Codes: F12, F13

Keywords: dumping, anti-dumping, counter-cyclical policy

<sup>&</sup>lt;sup>1</sup>Department of Economics, Georgetown University and Economic Research, Federal Reserve Bank of Chicago, 230 S. LaSalle St., Chicago, IL 60604; phone: (312) 919-5856; email: crowley.meredith@gmail.com. The views expressed here are those of the author and do not necessarily reflect those of the Federal Reserve Bank of Chicago or the Federal Reserve System. I thank Gadi Barlevy, Chad Bown, Bob Feinberg, Eric French, Anna Paulson, Tom Prusa, Rich Rosen and seminar participants at the Chicago Fed, the MWIEG Spring 2008 meetings, the Econometric Society NASM 2008, Georgetown, the Federal Reserve System Committee on International Economic Affairs, the US International Trade Commission, and the World Bank for helpful comments. Adam Hogan, Saad N. Quayyum and Erik Vogt provided excellent research assistance.

## 1 Introduction

Under the GATT-WTO system, countries which have negotiated reductions in tariffs and other trade barriers have limited ability to institute new trade barriers. However, the GATT includes provisions which allow countries to reintroduce trade barriers if specific economic criteria are met. Antidumping duties, permitted under the GATT's agreement on dumping, have become extremely popular among WTO members around the world (Miranda, Torres and Ruiz, 1998). According to the GATT, antidumping duties may be imposed if a country's trading partners are selling products at "dumped" prices and if there is evidence that the country's domestic industry is injured by this dumping.

This paper tests the theory that weak economic conditions in a foreign economy can induce a particular type of dumping prohibited by the GATT, cyclical dumping or temporarily pricing below the average total cost of production.<sup>2</sup> Models of cyclical dumping (Ethier, 1982; Staiger and Wolak, 1992; Crowley, 2007) predict that declines in the demand for a product in a foreign market will lead to increased exports sold at a price in the importing country that is below the average total cost of production. Using country-specific information on foreign economic shocks in manufacturing industries, I estimate a joint model of filing decisions by the US industry and antidumping decisions by the US government. I find that a decline in economic conditions in a foreign industry has a statistically and economically significant effect on the filing of antidumping petitions and the government's decision to impose antidumping measures. After controlling for other political and economic factors that likely drive industry filing and government decisions, including US GDP growth, I find that a one standard deviation fall in the growth of employment (consumption) in a foreign economy's manufacturing industry doubles (triples) the joint probability that the US industry will file an antidumping petition and the US government will impose a preliminary (temporary) antidumping measure. The effect of weakness in a foreign economy is even larger for final antidumping measures. A one standard deviation fall in foreign employment (consumption) growth increases the joint probability that a petition will be filed and a final (longlasting) antidumping measure will be imposed by a factor of five (seven).

Because even temporary, preliminary antidumping measures can have trade-restricting effect, I estimate models of both preliminary and final outcomes. In a binary model of the government's

<sup>&</sup>lt;sup>2</sup>Clarida (1996) presents estimates from a variety of sources that the definition of dumping as pricing below the average total cost of production is used in about 2/3 of US antidumping cases.

preliminary protection decision with industry selection into the antidumping process, the growth rate of employment in the foreign industry affects the government's decision to apply a preliminary antidumping measure. Conditional on an industry filing a request for antidumping protection, a one standard deviation fall in the growth of industry-level foreign employment in the year prior to the filing of the petition is associated with a modest increase in the probability of a preliminary measure of 0.74 percentage points. More importantly, a one standard deviation fall in foreign employment growth increases the probability of filing (0.23 percentage points) and, thus, more than doubles the joint probability that a petition will be filed and a preliminary measure will be imposed from 0.14% to 0.32%.

An interesting difference arises in estimating the role of foreign demand shocks on the government's final determination of whether or not to impose a final measure jointly with the industry's decision to file a petition. In the final decision, which takes place between 235 and 295 days after the filing of a petition<sup>3</sup>, the lagged growth of industry-level foreign employment is not a statistically significant determinant of the government's decision to impose a final long-lasting antidumping measure. However, because foreign economic conditions are a significant determinant of industry filings, a one standard deviation fall in foreign employment growth dramatically increases the joint probability of a filing and a final measure from 0.09% to 0.52%

This is the first paper that I am aware of to show empirically that US trade policy responds to adverse economic fluctuations in a foreign economy. This paper shares similarities with earlier work by Knetter and Prusa (2003) on the macro-economic factors underlying antidumping filing in that both papers exploit inter-temporal variation to identify the relationship between economic fluctuations and antidumping. Consistent with their findings and those of Feinberg (2005) and Jallab, Sandretto, and Gbakou (2006), I find that the probability of filing is increasing with an appreciation of the importing country's real exchange rate. However, my analysis differs from Knetter and Prusa (2003) by utilizing country and industry-specific inter-temporal variation to explain differences in filing behavior across accused countries and industries. While Knetter and Prusa found no statistically significant relationship between filing rates aggregated across industries and across accused countries and an aggregate measure of rest-of-world GDP growth over the 3 years prior to filing, my analysis finds that higher frequency, annual fluctuations in the foreign

 $<sup>^{3}</sup>$ The final dumping determination takes place between 235 and 295 days after the filing of a petition in normal cases, but can be delayed to a time range of 285 to 345 days in special circumstances. The final injury decision takes place 280-340 days after the filing of a petition in normal cases and 330-390 days in special circumstances.

economy's industry in the year prior to filing are an important determinant of filing.

Among theoretical models of dumping, the literature has developed into two strands that yield opposing predictions about the relationship between economic fluctuations in the foreign economy and the existence of dumping. Beginning with the seminal contribution of Ethier (1982), a number of papers (Staiger and Wolak, 1992, 1994; Crowley, 2007) model a realization of weak foreign demand as the driving force behind dumping. In contrast, Clarida (1993) develops a competitive model of entry and firm learning about technology to show that dumping can arise during periods of high worldwide demand. Similarly, Hartigan (1996) builds a duopoly model in which dumping occurs when foreign demand is high. The contribution of the current paper is that provides empirical support to models in which weak foreign demand drives dumping.

Previous empirical research on the determinants of antidumping filings and the outcomes in investigations has emphasized political factors (Hansen, 1990; Moore, 1992; Hansen and Prusa, 1997), specific aspects of the legal/bureaucratic institutional framework (Hansen and Prusa, 1996; Blonigen, 2006), trade policy retaliation (Blonigen and Bown, 2003<sup>4</sup>; Prusa and Skeath, 2002; Feinberg and Reynolds, 2006) or economic factors (Moore, 1992; Baldwin and Steagall, 1994; Staiger and Wolak, 1994; and Knetter and Prusa, 2003). The approach here builds on the previous empirical literature but utilizes a previously unexploited source of variation to identify a relationship between foreign economic fluctuations and dumping.

Section 2 describes the antidumping process in the United States. Section 3 presents the empirical model. Section 4 describes the data. Section 5 presents the empirical results and section 6 concludes.

## 2 Antidumping in the United States

This paper attempts to use information on the antidumping process in the United States over the period 1980-2001 to identify the role that industry-level cyclical economic factors in foreign economies play in dumping. In the course of an investigation that determines the existence and

<sup>&</sup>lt;sup>4</sup>Blonigen and Bown (2003) share a methodological similarity to this paper in that they exploit cross-country variation in petition filing to identify the effect of the threat of retaliation measured as foreign market size on filing decisions. Because their analysis utilizes cross-sectional data, they abstract away from the cyclical factors that theory predicts could explain time-variation in dumping behavior by foreign firms.

domestic consequences of dumping, the US government collects and analyzes private business proprietary data on prices and costs of domestic firms who make accusations of dumping and foreign firms accused of dumping. The government also analyzes a variety of publicly available data on output, imports and domestic employment that inform its decision in antidumping cases. Unfortunately, the econometrician who is interested in testing the theories of cyclical dumping cannot observe the business proprietary data on prices and costs that the government observes.

Nevertheless, I can observe industry and government decisions at several stages in the antidumping process. Furthermore, I have several measures that can proxy for the strength of demand in a foreign country's industry. The growth the growth of employment and the growth of consumption in the foreign country should all be highly correlated with the strength of demand in industry i in country j. According to cyclical dumping models, weakness in any of these variables in industry i, country j in year t induces (1) pricing below average total cost and (2) injury to the domestic import-competing industry (i.e., increased imports, lower domestic market share, lower domestic capacity utilization). Because weakness in foreign demand precipitates conditions that satisfy the legal definition of dumping, it should be associated with an increase in the probability of a petition by the domestic industry and an increase in the probability of protection.

Figure 1 presents, by antidumping petition status, the mean growth of foreign industry-level employment in the year before an antidumping petition is filed. This is the mean growth rate over 28 3 digit ISIC manufacturing industries for 49 countries between 1980-2001.<sup>5</sup> Figure 2 presents the mean growth of foreign industry-level consumption in the year prior to a filing. Beginning with figure 1, it appears that negative employment growth in a foreign industry is correlated with an antidumping case being filed and is also correlated with the imposition of a preliminary antidumping duty. Figure 2 shows that cases tend to be filed and preliminary duties tend to be imposed against foreign industries whose consumption growth is relatively weak. Somewhat surprisingly, the pattern appears reversed for final antidumping duties. In brief, this simple figure suggests that foreign is to use cross-industry cross-country and inter-temporal variation in antidumping decisions to identify if antidumping protection is induced by weak foreign demand and, if so, to quantify the increase in the probability of protection associated with economic weakness in foreign countries.

Identification of the effect of foreign demand strength on filings and outcomes comes from three sources of variation in the domestic-industry-foreign-country-year (ijt) panel data. First, within an

<sup>&</sup>lt;sup>5</sup>The data used in these graphs are described in detail in section 4.

industry i in year t, there is variation across countries j both in whether or not they are included in an antidumping petition and in the government's determination of injury and dumping. Second, within a year t, there is variation across industries i in filing and protection decisions. Finally, there is intertemporal variation for industry i and country j in the timing of petition-filing by industries and the protection decisions by governments.

The US antidumping process can be broken into three distinct phases, each with its own data requirements: 1. initiation of a case, 2. a preliminary phase with a determination as to whether or not dumped imports are causing or threatening to cause injury to the domestic industry and a preliminary determination about the existence and magnitude of dumping, and 3. a final phase with determination about the existence and magnitude of dumping and a final injury determination. <sup>6</sup>

In the first step, a domestic industry *i* that produces a product (also denoted  $i^7$ ) must initiate or file a petition with the US Department of Commerce (DOC) and the US International Trade Commission (USITC) that claims that the industry is being materially injured or threatened with materially injury by reason of imports of good *i* at a price that is "less than fair value" from a single country *j* or multiple countries j = 1, 2, 3... Thus, the first margin of cross-country within-industry and within-year variation comes in the filing decision.<sup>8</sup>

Table 1 presents the frequency of filing a petition in industry i against country j in year t for the 272,639 industry-country-year observations in the unbalanced panel of 439 industries, 49 countries and 22 years used in this paper. The first important observation is that antidumping petitions are

<sup>7</sup>In practice, the definition of the domestic "like product" is part of the investigation process. In general, the USITC definition of the "like product" is only a subset of the output of the 4 digit SIC 1987 industries that are the most finely disaggregated units available for a cross-sectional analysis in the US.

<sup>&</sup>lt;sup>6</sup>A detailed description of the current antidumping process and a brief history of the evolution of US antidumping law since 1916 can be found in the *Antidumping and Countervailing Duty Handbook* published by the US International Trade Commission (2007). Although US trade law has been revised a number of times since the start of my data sample in 1980, the antidumping process is largely unchanged with one important exception. Beginning in 1984, the rule to cumulate imports from all countries listed in a petition during the injury investigation led to super-additivity in USITC decisions and a 20-30% increase in the probability of protection (Hansen and Prusa, 1996). However, as Hansen and Prusa (1996) note, "the ITC always makes its decisions on a country-by-country basis, even if imports from a set of countries are cumulated."

<sup>&</sup>lt;sup>8</sup>Prusa (1997) and Benton (2001) exploit this variation to show that the country-specificity of antidumping protection leads to increased imports into the US (Prusa) and EU (Benton) from countries that are not specifically targeted by the antidumping measure.

a rare event. The row "Country j exports i to the US at t and is accused of dumping" indicates that the frequency of dumping accusations in the sample is less than 0.2% (545/272639). Second, the row "Country j exports i to the US at t and is not accused" captures the high degree of crosscountry variation in petition filing. There are 8017 observations in the dataset in which a country j with positive exports to the US in industry i year t was not listed in a antidumping petition filed by the domestic US industry i in year t. Thus, while only 545 petitions were filed in the unbalanced manufacturing panel, an additional 8017 petitions would have been filed if the domestic industry had indiscriminately listed every country j that exported industry i's output to the US in year t. It thus appears that domestic industries are somewhat selective in their accusations of dumping.<sup>9</sup>

This selectivity is likely influenced by the economic and financial data which are included in an industry's antidumping petition. In addition to data on the domestic industry itself (capacity, production, domestic sales, export sales, inventories, the number of production and non-supervisory workers, and income and loss data), a petition must include data on the foreign firms and countries accused of dumping. First, it must include the quantity and value of "less-than-fair-value" imports from each supplying country for the most recent 3 years. Second, it must include data directly related to the determination of dumping; i.e., the price charged by foreign firms in the US for the "dumped" product and the domestic firms' price for the same product for the most recent 5 quarters.<sup>10</sup>

If a petition contains all the necessary information, the preliminary phase of the investigation begins. During the preliminary phase, the US International Trade Commission makes a preliminary determination of injury and/or threat of injury due to dumped imports for each country jlisted in the petition. In making its decision, the USITC relies on information obtained through questionnaires sent to domestic producers, importers and foreign producers.<sup>11</sup> If the preliminary injury determination by the USITC is negative, the case ends with no antidumping duty. If the preliminary injury determination is affirmative, then the case proceeds to the US Department of

<sup>&</sup>lt;sup>9</sup>Blonigen and Bown (2003) explain this selectivity as partially due to a US industry's fear of foreign retaliation. <sup>10</sup>The International Trade Administration Form ITA-357P (OMB Control # 0625-0105) provides a detailed guide-

line of the price and cost information that must be included in a petition for antidumping protection.

<sup>&</sup>lt;sup>11</sup>Domestic producers' questionnaires request economic data on capacity, production, inventories, commercial shipments, export shipments, internal consumption, company transfers, employment, hours worked, wages, as well as financial data on income, losses, capital expenditures and sales prices. Importer questionnaires request data on the quantity and value of imports and sales prices. Finally, foreign producers' questionnaires inquire about the firm's capacity, production, home-market shipments, exports, and inventories.

Commerce for a preliminary dumping determination.<sup>12</sup>

During the second part of the preliminary phase, the US Department of Commerce makes a preliminary determination of dumping against each country j listed in a petition.<sup>13</sup> The dumping margin for country j is calculated as the difference in the price charged by country j producers in the US and country j,the US and a third market, or the difference between the price charged in the US price and the average total cost of production of a firm in country j. Preliminary dumping margins vary considerably across countries listed within a petition, with the USDOC occasionally finding that there is no evidence of dumping by some countries that have been found guilty of causing injury.<sup>14</sup>

To summarize the preliminary phase, a preliminary antidumping measure is imposed if both the USITC and the USDOC come to affirmative preliminary determinations. Cross-country variation in the application of a preliminary antidumping duty within a case can arise from cross-country variation in the injury decision or in the dumping determination. Interestingly, when petitions against individual countries are clustered into multicountry cases, we observe that roughly 73% of multi-country cases have outcomes in which all countries were found guilty of dumping and injury or no countries were found guilty of both dumping and injury. In the other 26% of cases, at least one country had a preliminary outcome that differed from that of the other countries accused in the multi-country case.

Table 2 provides evidence of this cross-country variation in preliminary outcomes for antidumping cases filed by US manufacturing industries between 1980 and 2001. The preliminary outcome is defined as affirmative if the US ITC finds preliminary evidence of injury or threat of injury and the USDOC finds preliminary evidence of dumping. The preliminary outcome is defined as negative if either agency makes a negative determination. The first column reports the number of countries

<sup>&</sup>lt;sup>12</sup>Although it is uncommon, the USITC occasionally splits its preliminary decision across countries. For example, in the 1993 case of phthalic anhydride (731-TA-664-668), only one of the five countries accused of dumping was found to be injuring the domestic US industry during the preliminary investigation. Similarly, in the 1995 case of polyvinyl alcohol (731-TA-726-729), only three of the four countries accused of dumping were found to be a cause of injury during the preliminary investigation.

<sup>&</sup>lt;sup>13</sup>This determination is made within 115 days of the USITC's preliminary determination. Therefore, the total elapsed time from initiation to a preliminary measure is 160 days in normal cases.

<sup>&</sup>lt;sup>14</sup>See, for example, the splits in preliminary dumping determinations in oil country tubular goods in 1994 (731-TA-711-717), stainless steel round wire in 1997 (731-TA-781-786), low enriched uranium in 2000 (731-TA-909-912), and structural steel beams in 2001 (731-TA-935-942).

named in an antidumping case, the second column lists the number of multicountry-cases, the third column lists the number cases in which all the accused countries faced a preliminary antidumping measure, and column 4 lists the number of multicountry cases in which no accused country faced a preliminary antidumping measure. Column 5 reports the number of cases in which some accused countries faced preliminary measures and some did not. Overall, in about 26% of antidumping cases at least one country had a preliminary outcome that was different from its accused peers.

If the preliminary injury investigation found evidence of injury caused by "dumped" imports, the case proceeds to the final phase. In the final phase, the USITC again sends questionnaires to the domestic firms, importers, and foreign firms that reported production and/or imports during the preliminary phase. Final questionnaires request generally the same data as preliminary questionnaires, but are updated with more recent data. One difference with the preliminary investigation is that questionnaires are sent to purchasers of the product requesting data on the value of purchases of the product manufactured domestically, by foreign firms accused of dumping and by other foreign firms. Purchasers are also asked to compare foreign and domestic products in terms of price, quality, service, delivery, etc. The US Department of Commerce makes a final dumping determination according to the price and cost data available. As in the preliminary phase, there is considerable variation across countries in the magnitude of the final duty with Commerce finding no evidence of dumping in some cases. Differences in the magnitudes of preliminary and final dumping margins can arise because more recent data on prices and costs have become available or because data obtained in the final questionnaires may obtain revisions to data obtained in the preliminary questionnaire.

To conclude the final phase of the investigation, the US ITC makes a final injury determination.<sup>15</sup> The USITC generally groups all countries j that have proceeded to the final phase together in its injury decision, but split decisions are possible if some countries are being investigated for a threat of injury (as opposed to actual injury). If there is evidence of injury and dumping, the government imposes a final antidumping duty.<sup>16</sup> In summary, during the final phase of an inves-

<sup>&</sup>lt;sup>15</sup>This determination is made within 120 days of the US DOC's preliminary dumping determination or 45 days of its final dumping determination, whichever date is later. Consequently, the total elapsed time from initiation of a petition to the final determination can range from 280 to 340 days in normal cases.

<sup>&</sup>lt;sup>16</sup>See Blonigen and Haynes (2002) and Blonigen and Park (2004) for a detailed discussion of the dynamics of antidumping duties after imposition. Since 1995, GATT rules limit the duration of antidumping measures to 5 years. However, Cadot, de Melo, and Tumurchudur (2007) find that US compliance with the WTO's sunset review policy is weak at best and likely has had no impact on the duration of US antidumping duties.

tigation, cross-country variation in which countries ultimately face antidumping duties can arise from variation in the outcome of the dumping determination or the injury (or threat of injury) determination.

Table 3 presents evidence on the cross-country variation in final determinations for antidumping cases brought by US manufacturers between 1980 and 2001. The structure of the table is the same as table 2. Again we observe that in the final stage of an antidumping investigation, the government split its decision across countries accused of dumping in one quarter of multi-country cases.

## 3 Empirical Model

To evaluate if foreign economic fluctuations lead to cyclical dumping, I estimate a binary model with selection (Van de Ven and Van Praag (1981)), specifically, a binary model of industry i petitions for antidumping protection against countries j and a binary model of the government's decision to impose antidumping measures. The empirical model is a two stage process. In the first stage, in every period t an industry i makes a binary decision to file for protection or not to file against each foreign country j that exports i's product to the US. In the second stage, if an industry has filed for protection, the government makes a binary decision of whether or not to impose an antidumping measure against each country j accused of dumping.

To simplify the analysis, in the second stage, I estimate a binary model of the government's decision to impose a preliminary antidumping measure in which the injury determination by the USITC and the dumping determination by the USDOC are collapsed into a single outcome variable which is affirmative if both agency decisions are affirmative. Similarly, the binary model of the final decision is collapsed into a single outcome variable which is affirmative if both agency decisions are affirmative if affirmative if both dumping and injury.

In the second stage, the government's latent measure of injury and dumping  $d_{ijt}^*$  is unobserved, but takes the form  $d_{ijt}^* = \beta' x_{ijt} + \varepsilon_{ijt}$  where *i* denotes the industry in which dumping is alleged to occur, *j* denotes the foreign country accused of dumping, and *t* denotes the time period in which the complaint is filed. The variables in  $x_{ijt}$  are described in detail in the next section. In brief, this vector includes a measure of the state of industry demand in both the accused foreign country and in the importing country and lagged measures of injury to the importing country's industry. Although I do not observe the latent measure of injury and dumping, I observe the importing government's decision of whether  $(d_{ijt} = 1)$  or not  $(d_{ijt} = 0)$  to impose antidumping protection conditional on an industry filing for protection.

$$d_{ijt} = \begin{cases} 1 & \text{if } d_{ijt}^* > 0 \\ 0 & \text{if } d_{ijt}^* \le 0 \end{cases}$$
(1)

Assuming  $\varepsilon_{ijt} \sim N(0,1)$ , then the likelihood for the selected sub-sample is

$$L = \prod_{ijt} \left[ \Phi(\beta' x_{ijt}) \right]^{d_{ijt}} \prod_{ijt} \left[ 1 - \Phi(\beta' x_{ijt}) \right]^{(1-d_{ijt})}$$
(2)

where the CDF for a normal distribution is denoted by  $\Phi$ .

An antidumping case is only considered by the government if a domestic industry chooses to file a petition for protection. If an industry's decision to apply for protection and the government's decision to grant protection are correlated, then estimates of  $\beta$  will be inconsistent.

In the first stage, the industry's latent measure of selection,  $y_{ijt}^*$ , is unobserved, but takes the form  $y_{ijt}^* = \gamma' z_{ijt} + \nu_{ijt}$ , where  $z_{ijt}$  is a vector that includes a measure of the state of industry demand in the foreign country and foreign and domestic industry characteristics that are predetermined at time t;  $(\nu_{ijt}, \varepsilon_{ijt})$  are assumed to be independent of  $(x_{ijt}, z_{ijt})$  and their joint distribution is given by:

$$\begin{bmatrix} \varepsilon_{ijt} \\ \nu_{ijt} \end{bmatrix} \sim N\Big(0, \begin{bmatrix} 1 & \rho \\ \rho & 1 \end{bmatrix}\Big)$$

The industry's decision to petition  $(y_{ijt} = 1)$  can be written

$$y_{ijt} = \begin{cases} 1 & \text{if } y_{ijt}^* > 0 \\ 0 & \text{if } y_{ijt}^* \le 0 \end{cases}$$
(3)

Given the preceding assumptions, the expectation of the government's latent variable in the second stage can be written:

$$E(d_{ijt}^{*}|x_{ijt}, y_{ijt}^{*} > 0) = E(\beta' x_{ijt}|x_{ijt}, \nu_{ijt} > -\gamma' z_{ijt}) + \rho \frac{\phi(-\gamma' z_{ijt})}{\Phi(\gamma' z_{ijt})}$$
(4)

and the government's latent variable is given by:

$$d_{ijt}^* = \beta' x_{ijt} + \rho \frac{\phi(-\gamma' z_{ijt})}{\Phi(\gamma' z_{ijt})} + \tilde{\varepsilon}_{ijt}$$
(5)

where  $\tilde{\varepsilon}_{ijt}$  is normally distributed and independent of  $\nu_{ijt}$ ,  $E(\tilde{\varepsilon}_{ijt}|y_{ijt}^*>0) = 0$  and  $E(\tilde{\varepsilon}_{ijt}^2|y_{ijt}^*>0) = 1 - \rho^2 \lambda_{ijt}(-\gamma' z_{ijt} - \lambda_{ijt})$  and where  $\lambda_{ijt} = \phi(-\gamma' z_{ijt})/\Phi(\gamma' z_{ijt})$ .

Renormalizing  $d_{ijt}^*$  so that the variance of the censored error,  $\tilde{\varepsilon}_{ijt}$ , is equal to one allows us to derive the likelihood for the full model as:

$$L = \prod_{ijt} \left[ \Phi(\beta' x_{ijt}, \gamma' z_{ijt}, \rho) \right]^{d_{ijt} y_{ijt}} \prod_{ijt} \left[ \Phi(-\beta' x_{ijt}, \gamma' z_{ijt}, \rho) \right]^{(1-d_{ijt}) y_{ijt}} \prod_{ijt} \left[ \Phi(-\gamma' z_{ijt}) \right]^{1-y_{ijt}}$$
(6)

Identification of the effect of foreign demand strength on filings and outcomes comes from three sources of variation in the ijt panel data. First, within an industry i in year t, there is variation across countries j both in whether or not they are included in an antidumping petition and in the government's determination of injury and dumping. Second, within a year t, there is variation across industries i in filing and protection decisions. Finally, there is intertemporal variation for industry i and country j in the timing of petition-filing by industries and the protection decisions by governments.

Marginal effects and predicted joint probabilities derived from coefficient estimates obtained from maximizing the log of the likelihood (6) are reported in tables 6 - 13. The maximum likelihood coefficient estimates are reported in appendix tables C- F.

#### 4 Data

I estimate the empirical model using a panel dataset constructed from three main data sources: (1) the World Bank's Trade, Production and Protection Data, (2) the NBER Trade and Manufacturing Databases, and (3) the Global Antidumping Database maintained by Chad Bown. Data on US GDP growth come from the US Bureau of Economic Analysis. Annual bilateral real exchange rates in foreign currency per US dollar come from the USDA Economic Research Service. Summary statistics for all variables in the dataset are reported in table 5.

The focus of the empirical work is quantifying the role that foreign demand shocks play in industry petitioning and the government's decision to impose antidumping protection. The World Bank's Trade, Production and Protection Data compiled by Alessandro Nicita and Marcelo Olarreaga provides information on output, imports, exports and employment for 28 3 digit ISIC Rev. 2 manufacturing industries from 1980-2001 for 49 developed and developing economies.<sup>17</sup> Because US antidumping duties are imposed against countries and not customs unions or free trade areas, the analysis treats each country of the EU as a separate observation with country-level, rather than Union-level data used in the analysis.

Estimation of the model uses four different proxies for the strength of demand in country j industry i: the growth of employment for ijt - 1, the growth of consumption (output plus imports less exports) for ijt-1, detrended employment growth and detrended consumption growth. Although foreign consumption growth would seem to be the best measure to capture fluctuations in foreign demand, the rationale for focusing on foreign employment growth is that we might expect the reported counts of workers to be a relatively high quality variable across a wide spectrum of countries. Moreover, data on industry employment is available for more countries and years. The two detrended growth measures are intended to better capture country and industry-specific demand fluctuations. The detrended variable is defined as employment (consumption) growth in country j industry i at t - 1 less the average annual growth rate of employment for country j industry i from 1979 (or earliest year available) until 2001. The logged levels of employment and consumption serve as proxies for the size of the foreign industry. Because the current value of these variables could be endogenous to filing behavior by US industries, I use the first lag of all foreign variables in estimating the model.

The NBER Trade and Manufacturing Databases provide data on imports, shipments, prices, employment, real capital stock and value added for about 450 manufacturing industries. These data were concorded to 439 1987 4 digit SIC codes using the industry concordance provided by the NBER-CES Manufacturing Industry Database and the original data from 1979-1994 were extended through 2001. Nominal values of imports and shipments (a measure of domestic output) were

<sup>&</sup>lt;sup>17</sup>Countries in the dataset include: Argentina, Australia, Austria, Bangladesh, Belgium, Brazil, Canada, Chile, China, Columbia, Costa Rica, Denmark, Ecuador, Egypt, El Salvador, Finland, France, Germany, Greece, Hong Kong, Hungary, India, Indonesia, Ireland, Israel, Italy, Japan, Kenya, South Korea, Malaysia, Mexico, Netherlands, New Zealand, Norway, Peru, Philippines, Poland, Portugal, Singapore, South Africa, Spain, Sweden, Switzerland, Taiwan, Thailand, Trinidad, Turkey, United Kingdom, and Venezuela. Twenty-two years of disaggregated manufacturing data are available for almost all of the countries in the dataset. Important targets of US antidumping duties with fewer than twenty-two years of available data include: the People's Republic of China (1980-1998), Germany (1992-2001), and Brazil (1993-1996).

deflated to real 1987 dollars using industry specific price indices.

Industry characteristics used to estimate the selection equation include political and economic measures that may affect an industry's propensity to file but are thought to be unrelated to the government's determination of dumping. Some industries may be more likely to file for protection than others. For example, large industries may be better able to assume the large legal fixed cost of filing a petition. Industries in which the level of imports relative to total domestic consumption is high may be more familiar with trade protection policies and thus, more likely to file. The vertical structure of an industry may matter; industries that are further downstream may file more petitions because they are more sensitive to industry price changes. Thus, a measure of industry size, the level of employment; the real import penetration ratio (real imports/(real imports + real domestic shipments)); and a proxy for the vertical structure of an industry, the value-added to output ratio are used to estimate the selection equation. The selection equation also includes three measures of injury which US law suggests should be important to the government's decision; the capacity utilization rate (real shipments/real capital stock), the percent change in the import penetration ratio and the change in employment. Because the current values of industry specific variables and the choice of whether to petition for protection may be endogenous, I use lagged values of these variables in  $z_{iit}$ .

Data on antidumping cases from 1979 through 2001 come from Global Antidumping Database Version 3.0 compiled by Chad Bown at Brandeis University. The US files in the database provide detailed information on the date a petition was filed, the identity of the country accused of dumping, tariff line information on the products involved, various outcome dates, and the outcomes at different stages of the investigation. The variable on petition filings takes on a value of one if the US industry filed a petition. Using the detailed information on the case outcome, I construct two binary measures of the government's determination: preliminary decision and final decision. Following Hansen and Prusa (1995), the preliminary (final) decision is coded as affirmative if the US government imposed a preliminary (final) antidumping duty or if he case resulted in a suspension or termination agreement during the preliminary (final) phase.<sup>18</sup>

<sup>&</sup>lt;sup>18</sup>In appendix tables A (preliminary duties) and B (final duties) I report results in which suspension and termination agreements were re-coded as negative (no-duty) outcomes. The estimated marginal effects of foreign economic conditions are similar across coding schema although the magnitudes are larger when suspensions and terminations are coded as no-duty outcomes. The robustness across coding schema is likely due to the rarity of suspensions and terminations after the late 1980s.

The three datasets used in the paper provide information on industries at three different levels of aggregation: 3 digit foreign industries, 4 digit US industries, and 5, 7, 8, or 10 digit antidumping *cases.* In order to merge these three datasets together, the 439 4 digit SIC87 manufacturing industries were mapped into 28 ISIC R. 2 manufacturing industries by the author. Similarly, the tariff-line level antidumping case data were also mapped into 439 4 digit SIC87 industries by the author. Because the US and foreign industry data used in the analysis are more aggregated than the industries investigated in antidumping cases, we might expect parameter estimates based on these aggregated variables to underestimate the true effect of domestic and foreign factors on antidumping outcomes.

## 5 Empirical Results

The empirical results summarized in tables 9 and 13 indicate that a weakening of demand in a foreign industry is associated with an increased probability of US antidumping protection. According to the estimates of the empirical model, a one standard deviation fall in foreign demand increases the joint probability that a US industry will seek and the US government will grant a preliminary antidumping measure by a factor of two to three, depending on the measure of foreign demand used. Similarly, a one standard deviation fall in foreign demand increases the joint probability of a filing and a final antidumping measure by a factor of five to seven, depending upon the precise measure. This suggests that the US government might be using antidumping protection to shelter the US economy from cyclical dumping caused by economic weakness in a foreign trading partner.

Tables 6 - 8 report the estimated marginal effects on the probability that an industry i will file an antidumping petition against country j in year t and in the probability that the US government will impose a preliminary antidumping measure. The predicted joint probability of a petition filing and a preliminary antidumping measure associated with a one standard deviation fall in foreign demand is presented in table 9.

Tables 10 - 12 present estimated marginal effects for petition filings and final antidumping measures. Table 13 summarizes these results as the predicted joint probability of a petition filing and a final antidumping measure in response to a one standard deviation fall in foreign demand. Maximum likelihood coefficients for all variables are reported in appendix tables C - F.

#### 5.1 Preliminary antidumping measures

Table 6 presents estimates of the two-stage model of the US industry's filing decision and government's decision to impose a preliminary antidumping measure. The top panel of table 6 reports the marginal increase in the probability that a US industry will file an antidumping petition associated with changes in measures of foreign demand and the size of the foreign industry. The bottom panel reports the marginal increase in the probability of the US government imposing a preliminary antidumping measure.

Column 1 reports results for the basic specification. A one unit increase in the lagged growth of foreign employment is associated with a 1.2% decrease in the probability that a domestic industry will file an antidumping petition. Evaluated at the sample mean, a one standard deviation fall in lagged foreign employment growth increases the probability that that country's industry will face an antidumping petition by 0.23 percentage points. This roughly doubles the probability of a filing relative to the unconditional mean filing rate in the sample of 0.20%.

Proceeding to the lower panel of the table, in column 1 a one-unit increase in the growth of foreign employment is associated with a reduction in the probability of an preliminary antidumping measure of 3.8% after controlling for other factors (the lagged employment of the foreign industry, the lagged growth of country-specific import penetration, lagged US GDP growth, and the lagged bilateral exchange rate) and the domestic industry's first-stage decision to file an antidumping petition. The economic significance is moderate; a one standard deviation fall in foreign employment growth translates into a 0.74 percentage point increase in the probability of an affirmative preliminary decision. Combining estimates from the two stages of the model, the predicted joint probability of a petition fall in lagged foreign employment growth is 0.32%, roughly 2 times the mean joint probability in the sample of 0.14%. Table 9 reports predicted joint probabilities for all specifications.

Moving across the columns of table 6, alternative measures of the strength of foreign demand are employed. Column 2 substitutes the lagged growth of foreign consumption for the lagged growth of employment. The results for the main variable of interest are of the same sign and similar magnitude in the industry filing equation, even though the sample size is reduced by roughly 50,000 observations when this less-available measure of foreign demand is used. On this smaller sample, the coefficient on lagged foreign consumption growth is not significantly different from zero in the government's decision equation. Table 9 reports that the predicted joint probability of a petition and a preliminary measure is 0.44% in response to a one standard deviation fall in lagged foreign consumption growth.

This raises a question: are the differences in the estimates presented in columns 1 and 2 due to differences in the measures of foreign demand or due to differences in the estimation samples. The results in column 3 attempt to resolve this. Column 3 repeats the specification 1 on the smaller sample of observations for which the lagged foreign consumption growth variable is available. On this smaller sample, the marginal effect of a change in foreign employment growth is larger in the industry's filing equation, and not significant in the government's decision equation. To assess the quantitative significance of these differences, table 9 reports that the predicted joint probability of a petition and a preliminary measure conditional on a one standard deviation fall in the two measures of foreign demand. For both specifications 2 and 3, the joint probability is 0.44%. This suggests that the differences in estimates across specifications are likely due to the changing sample size rather than in a fundamental difference in what the variables are measuring.

In specification 4, the measure of foreign demand is the actual growth of foreign employment in the year before the antidumping petition was filed less the long-run trend growth of foreign employment in country j's industry i. This detrended measure of foreign employment growth might better capture the country and industry-specific cyclicality of foreign demand. The results are close in magnitude to those of the basic specification. In table 9 the predicted joint probability of an antidumping measure for this specification is 0.29%, a hair smaller than the predicted probability from specification 1 of 0.32%. Specification 5 repeats this type of analysis by using detrended foreign consumption growth. Again, the estimates are quantitatively similar in size with a predicted joint probability of 0.39% compared to 0.44% from specification 2. Together, columns 4 and 5 indicate that the results are robust to different measures of foreign demand that might better capture country and industry specific cycles.

Appendix tables C and D report the maximum likelihood coefficients for all variables in the joint model of the government's protection decision and the industry's filing decision. The coefficient estimates of US variables (the growth of lagged import penetration, the lagged level of import penetration, the lagged growth of employment in the domestic industry, the lagged level of domestic employment, the lagged level of capacity utilization, the lagged growth of US GDP, and the lagged level of the bilateral real exchange rate) are generally in line with previous research. Interestingly, coefficient estimates reported in appendix table C on the lagged import penetration ratio and the lagged growth of import penetration in the selection equation tell us that (1) consistent with Staiger and Wolak's (1994) findings, industries with higher import penetration ratios are more likely to seek protection and further, (2) countries with higher import penetration ratios are more likely to be targets of antidumping activity than countries with lower import penetration ratios. However, a shortcoming of the import penetration variable is that it cannot inform us as to whether industries seek (and the US government imposes) antidumping protection because US shipments have fallen or because foreign imports have risen. Thus, the finding that antidumping protection is more likely against a country that is experiencing relatively weak growth in its own industry after controlling for import penetration informs our understanding of why there is variation in the government's policy response to two different countries or industries with similar levels of import penetration. Consistent with the findings of Knetter and Prusa (2003), Feinberg (2005) and Jallab, Sandretto and Gbakou (2006) all of whom estimate negative binomial models of filing rates, a real appreciation of the dollar is associated with an increase in the probability of antidumping filings.

The second to last line of table 6 reports estimates of  $\rho$ , the correlation of the errors from the first and second stage. Estimates of  $\rho$  in table 6 are negative and of roughly the same magnitude across all specifications. A negative selection bias is somewhat surprising. One possible interpretation is that US industries only apply for protection if they are in poor health and think that they might reasonably satisfy the injury criteria. Among this set of poorly-performing industries, the government chooses to protect only the relatively healthy industries with an antidumping measure.

The last row of table 6 reports the loglikelihood. Because the inclusion of different variables changes the estimation sample, it is not feasible to use the loglikelihood to compare fit across all models. However, comparisons can be made between specifications 2 and 3, 1 and 4, and 3 and 5. The differences in model fit, as shown by the loglikelihoods, are small.<sup>19</sup>

Table 7 address the question of whether or not the results from table 6 are driven by particular industries or countries. Japan and China have been frequent country targets of antidumping protection<sup>20</sup> while steel and chemicals are the most cited industries. The estimates in table 7 add country and industry fixed effects to specifications 1-4 to assess the robustness of the results.<sup>21</sup> Es-

<sup>&</sup>lt;sup>19</sup>The AIC and BIC for all specifications are reported in appendix table C. As with the loglikelihood criterion, the differences in model fit according to the information criteria are small.

<sup>&</sup>lt;sup>20</sup>Previous studies (Bown, Crowley, McCulloch, and Nakajima, 2005; Bown and Crowley, 2007; and Bown, 2007) have documented that China's treatment by antidumping authorities is unique; either because of its non-marketeconomy status, because of the perceived threat of its relatively large export capacity, or for other reasons.

<sup>&</sup>lt;sup>21</sup>The incidental parameters problem means that the inclusion of country and industry fixed effect could lead to inconsistent estimates if number of fixed effects increases as the number of observations increases. In this application

timates in the fixed effects specifications 6-9 are of the same sign as the specifications without fixed effects (1-4) although the magnitudes of the marginal effects vary. Table 9 presents the quantitative results from table 7 as predicted joint probabilities. In specification 6, the inclusion of country and industry fixed effects yields a larger predicted joint probability of an antidumping measure of 0.51% in response to a one standard deviation fall in foreign employment growth compared to the predicted joint probability of 0.32% for the specification with no fixed effects (1). For consumption growth, specification 7 yields a smaller predicted joint probability of 0.23% when fixed effect are included. Specifications 8 (employment growth on a small sample) and 9 (detrended employment growth) yield predicted joint probabilities of 0.38% and 0.49%. The results from table 7 show that the results are not driven by specific countries or industries. In fact, the effect of a fall in foreign employment growth on the probability of an antidumping measure is larger when country and industry fixed effects are included.

Table 8 presents additional variations of the basic specification. In specification 10, the panel dataset is restricted to the 4 digit SIC87 industries that have filed at least one antidumping petition against any country j between 1980 and 2001. This specification relies heavily on time series variation for frequent industry users of antidummping to obtain identification. In table 9, a one standard deviation fall in foreign employment growth generates a predicted joint probability of a preliminary antidumping measure of 0.64%, suggesting that cyclical variation in the foreign economy might be even more important in industries that regularly seek antidumping protection. Specification 11 attempts to control for the fact that small exporters are not subject to antidumping duties.<sup>22</sup> This sample drops all country-product-year observations in which a country's share of the US import market is less than 1% of a 4 digit SIC87 industry. After omitting these smaller exporters, the joint probability of a preliminary antidumping measure is 0.92% conditional on a one standard deviation fall in foreign industry employment growth. This is more than twice as large as the joint probability with 22 years of data, there is one fixed effect for each of the 49 countries and there are 28 2 digit industry-level fixed effects for the 439 4 digit industries in the sample. Thus, as the number of fixed effects is far less than the number of observations, the extent of the inconsistency might not be too severe. However, the results should be viewed with this in mind.

<sup>22</sup>The WTO's antidumping code states that any country which is the source of less than 3% of the imports of a product that is subject to an antidumping investigation is a "negligible" supplier and cannot be subject to antidumping duties. Because my analysis uses industry definitions at the 4 digit SIC87 level which are more aggregated than the product definitions used in antidumping cases, I restrict the "large exporter" sample to countries with an import market share greater than 1% of the 4 digit industry. Results are robust to modest changes in this definition.

of a filing and preliminary measure in this "large exporter" sample of 0.44%. Finally, specification 12 seeks to identify if antidumping investigations that involve more than one country are more or less likely to face a preliminary antidumping duty that single-country investigations. The estimate on the multicountry case dummy is not statistically significant, suggesting that multicountry cases are not more likely to result in preliminary antidumping measures than single country cases after controlling for other factors. The marginal effect of a foreign demand shock has the same sign and a slightly larger magnitude in both stages of the non-linear model. The predicted joint probability of 0.41% is slightly larger than for the basic specification. Overall, table 8 validates the robustness of the basic results while suggesting that the magnitude of foreign economic shocks is likely larger for industries that regularly seek protection under antidumping law and for important exporting countries.

In summary, weak growth in either employment or consumption in foreign industries is a powerful predictor of a preliminary antidumping measure. After controlling for other economic and political factors including US GDP growth, a one standard deviation fall in the lagged growth of foreign demand increases the probability of a preliminary antidumping measure by a factor of 2 to 3, depending on the precise measure of foreign demand used. In conclusion, temporary US trade policies are applied counter-cyclically to foreign economic fluctuations.

#### 5.2 Final antidumping measures

Table 10 presents marginal effects for the model of industry filing and the government's decision to impose final antidumping measures. A one standard deviation fall in foreign employment growth has a significant positive effect on the probability that a petition will be filed, but is not a statistically significant determinant of the government's decision to impose final (long-lasting) antidumping measures. Quantitatively, the predicted joint probability of a petition and a final antidumping measure evaluated at the sample mean is 0.52% in response to a one standard deviation fall in foreign employment growth. This is a more than five-fold increase above the underlying joint probability in the sample of 0.09%. See table 13. Substituting in foreign consumption growth (specification 14), using a smaller sample (specification 15), or using detrended employment (specification 16) or consumption growth (specification 17) as the measure of foreign demand generate broadly consistent results. The strength of foreign demand is an important determinant of filing behavior, but not of the government's decision. In table 13, the predicted joint probabilities of a petition and a final measure are 0.69%, 0.56%, 0.48% and 0.49% for specifications 14-17, respectively. It is not clear why the measures of foreign demand are not significant determinants in the government's final decision. One possibility is that new data might become available to policy makers between the time of the preliminary and final decision. However, because the overwhelming majority of preliminary and final decisions are made within the same calendar year, there is not enough variation in the sample to examine this with annual data.<sup>23</sup>

As with preliminary measures, the coefficient estimates for other variables are generally in line with previous research. Maximum likelihood coefficient estimates for all variables are reported in appendix tables E and F. The estimated correlation coefficient is negative in all specifications.

Table 11 adds country and 2-digit industry fixed effects to four of the specifications in table 10. Across the board, the inclusion of the fixed effects leads to larger marginal effects of the measure of foreign demand on the probability of filing an final antidumping petition. This translates into large increases in the predicted joint probability of a petition filing and a final measure imposed. See table 13. A one standard deviation fall in foreign employment growth (specification 18) yields a predicted joint probability of 0.90%, a ten-fold increase over the sample's joint probability of 0.09%. Similarly, a one standard deviation fall in foreign consumption growth (specification 19) yields a predicted joint probability of 1.00%.

Finally, table 12 reports three addition specifications. Column 22 presents estimates obtained from a sample restricted to 4 digit SIC87 industries that filed at least one antidumping petition during the sample period. Again, the sign of the marginal effect of foreign employment growth is the same as that of the basic specification 13 reported in table 10. Quantitatively, table 13 reports a one standard deviation fall in lagged foreign employment growth increases the joint probability of a filing and a final measure to 0.71%. Specification 23 restricts the sample to large exporters as defined above. Within this sample of large exporters, a one standard deviation fall in lagged foreign employment growth increases the joint probability of a filing and a final measure to 0.45% from a mean in this sample of 0.27%. Specification 24 adds a multicountry case dummy to the government's decision rule and yields an estimate of the joint probability of a filing and a final measure associated with a one standard deviation fall in lagged foreign employment growth of 0.55%.

To summarize tables 10 - 12, economic weakness in foreign industries is a strong predictor of

<sup>&</sup>lt;sup>23</sup>Estimating the model with the data lagged from the date of the final decision rather than the petition filing yielded no notable changes. Because most cases are completed in one calendar year, this alternative lagging structure yielded few changes relative to the basic data sample.

final antidumping measures. However, the mechanism by which these variables influence the outcome is different for final (long-lasting) antidumping duties and preliminary (temporary) measures. While measures of economic weakness in a foreign industry directly impact both the government's preliminary antidumping decision rule and the domestic industry's filing rule, for final antidumping duties, measures of foreign demand only affect the ultimate outcome through the domestic industry's filing decision. Depending on the exact specification of the model, a one standard deviation fall in foreign demand can increase by a factor of 10 the joint probability that an industry will file an antidumping petition and the US government will impose a final antidumping duty.

#### 5.2.1 Robustness

In appendix tables A and B, I check the robustness of these results against an alternative definition of the dependent variables. Appendix table A (appendix table B) reports results in which the dependent variable is coded 1 if the US government imposed a preliminary (final) duty and 0 if the petition resulted in no duty or a suspension or termination agreement. This coding presumes that negotiated suspension and termination agreements have a less trade-restricting effect than duties.<sup>24</sup>

Specifications 25-29 re-estimate the preliminary investigation models of specifications 1-5 with the alternative dependent variable. In all specifications, the marginal effect of the foreign demand variable has the same sign but a larger magnitude when suspension and termination agreements are coded as a negative outcome.

For final antidumping investigations, the results are similar. Coding suspension and termination agreements as negative outcomes of the final antidumping process yields marginal effects of foreign demand in specifications 30-34 than are roughly equal or larger in magnitude that those in specifications 13-17.

The broad consistency of the results across coding schema is not surprising given the rarity of suspension or termination agreements in the sample that extends from 1980 through 2001.

 $<sup>^{24}</sup>$ Hansen and Prusa's (1988) work on the trade-restricting effects of non-duty outcomes like suspension and terminations agreement provided the support for coding suspension and termination agreements as affirmative antidumping measures in tables 6 - 13.

## 6 Conclusion

This paper exploits cross-country, cross-industry and intertemporal variation in manufacturing employment and consumption growth to identify the relationship between economic weakness in a foreign economy and the probability of antidumping protection. After controlling for other economic and political variables including US GDP growth, I find that a one standard deviation fall in the growth of employment (consumption) in a foreign economy's manufacturing industry doubles (triples) the joint probability that the US industry will file an antidumping petition and the US government will impose a preliminary (temporary) antidumping measure. Further, a one standard deviation fall in foreign employment (consumption) growth increases the joint probability that a petition will be filed and a final (long-lasting) antidumping measure will be imposed by a factor of five (seven). While earlier research by Knetter and Prusa (2003) showed that antidumping protection is applied counter-cyclically with regard to the domestic business cycle, this paper provides the first evidence that I am aware of that US trade policy is applied counter-cyclically to foreign economic fluctuations. Figure 1: Mean employment growth in industry i country j in the year before an antidumping petition is filed



Figure 2: Mean consumption growth in industry i country j in the year before an antidumping petition is filed



Table 1: Accusations of dumping	: US Manufacturing 1980	)-2001
---------------------------------	-------------------------	--------

Industry $i$ makes accusation of dumping at $t$				
Country $j$ exports $i$ to US at $t$ and is				
accused of dumping	545			
not accused	8017			
Industry $i$ makes no accusation of dumping at $t$				
Country $j$ exports $i$ to US at $t$ and is				
not accused	264,077			

No. of countries	Number	Affirm.	Negative	Split
named in case	of cases	Cases	Cases	Cases
1	186	138	48	0
2	65	43	10	12
3	18	8	5	5
4	14	8	5	1
5	8	5	0	3
6	4	3	0	1
7	5	2	1	2
8	1	0	0	1
12	1	0	1	0
Total	302	207	70	$\overline{25}$

Table 2: Cross-country variation in prelim. antidumping outcome: 1980-2001

No. of countries	Number	Affirm.	Negative	$\operatorname{Split}$
named in case	of cases	Cases	Cases	Cases
1	186	80	106	0
2	65	28	25	12
3	18	7	6	5
4	14	5	4	4
5	8	4	2	2
6	4	2	1	1
7	5	0	4	1
8	1	0	1	0
12	1	0	1	0
Total	302	126	150	25

Table 3: Cross-country variation in final antidumping outcome: 1980-2001

Table 4: 49 countries included in the dataset

Argentina	Costa Rica	Hungary	Mexico	Spain
Australia	Denmark	India	Netherlands	Sweden
Austria	Ecuador	Indonesia	New Zealand	Switzerland
Bangladesh	Egypt	Ireland	Norway	Taiwan
Belgium	El Salvador	Israel	Peru	Thailand
Brazil	Finland	Italy	Philippines	Trinidad
Canada	France	Japan	Poland	Turkey
Chile	Germany	Kenya	Portugal	U. Kingdom
China	Greece	South Korea	Singapore	Venezuela
Columbia	Hong Kong	Malaysia	South Africa	

	Mean	Std. Dev.
Dependent Variables		
Preliminary Measure= $1 $ petition filed	0.739	0.439
Final Measure= $1 $ petition filed	0.459	0.499
Petition Filed	0.0020	0.0447
Measures of foreign industry demand		
Growth Foreign $\text{Employment}_{ijt-1}$	0.003	0.195
Growth Foreign Consumption $_{ijt-1}$	0.053	0.283
Growth Foreign $\text{Employ}_{ijt-1}$ - $\text{Trend}_{ij}$	0.001	0.188
Growth Foreign $\text{Consump}_{ijt-1}$ - $\text{Trend}_{ij}$	0.003	0.275
Measures of foreign market size		
Ln Level of Foreign $\text{Employment}_{ijt-1}$	10.572	1.703
Ln Level of Foreign Consumption $_{ijt-1}$	14.867	1.778
Domestic Industry Variables		
Growth US $\text{Employment}_{it-1}$	-0.014	0.095
Ln US $\text{Employment}_{it-1}$	3.202	1.031
Capacity Utilization $_{it-1}$	2.765	1.743
$ValueAdded/Output_{it-1}$	0.505	0.119
Foreign*Domestic Industry Variables		
Growth Import Penetration $_{ijt-1}$	0.088	1.118
Import Penetration $_{ijt-1}$	0.006	0.026
Number of Observations	272639	

Table 5: Summary Statistics

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (2) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consumption growth as foreign demand shock
	(1)	(2)	(3)	(4)	(5)
<u>Stage 1: Industry's filing decision</u> Measures of foreign demand Growth of foreign employment, iit-1	-0 012***		-0 023***		
Growth of foreign consumption_ijt-1	(0.004)	-0.014**	(0.004)		
Detrended growth of foreign employment_ijt-1		(0.000)		-0.010** (0.004)	
Detrended growth of foreign consumption_ijt-1					-0.012** (0.006)
Measures of foreign market size Ln foreign employment_ijt-1	0.008*** (0.001)		0.009*** (0.001)	0.007*** (0.000)	
Ln foreign consumption_ijt-1		0.009*** (0.001)			0.009*** (0.001)
Other control variables	yes	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.038* (0.023)		-0.040 (0.031)		
Growth of foreign consumption_ijt-1		0.000 (0.015)			
Detrended growth of foreign employment_ijt-1				-0.042* (0.022)	0.000
Measures of foreign market size					-0.003 (0.015)
Ln foreign consumption ijt-1	0.003 (0.003)	0.002	0.004 (0.003)	0.003 (0.002)	0.002
0 1 _/		(0.004)			(0.004)
Other control variables	yes	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545	443
Number of Observations	272639	228830	227849	272639	228830
Rho	-0.025** (0.012)	-0.027* (0.015)	-0.029* (0.015)	-0.024** (0.011)	-0.027* (0.015)
Log Likelihood	-3616.14	-2952.25	-2937.34	-3616.09	-2953.45

#### Table 6: Estimates of marginal effects for preliminary decisions

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Add country and industry fixed effects to col. (1)	Add country and industry fixed effects to col. (2)	Add country and industry fixed effects to col. (3)	Add country and industry fixed effects to col. (4)
	(6)	(7)	(8)	(9)
Stage 1: Industry's filing decision				
Measures of foreign demand				
Growth of foreign employment_ijt-1	-0.025** (0.01)		-0.018*** (0.005)	
Growth of foreign consumption_ijt-1		-0.004** (0.002)		
Detrended growth of foreign employment_ijt-1				-0.024** (0.011)
Measures of foreign market size				
Ln foreign employment_ijt-1	0.019*** (0.002)		0.008*** (0.001)	0.018*** (0.002)
Ln foreign consumption_ijt-1		0.001*** (0.001)		
Other control variables	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand				
Growth of foreign employment lit-1	-0.104*		-0.044	
	(0.059)		(0.031)	
Growth of foreign consumption_ijt-1	, , ,	0.002 (0.006)	ζ, γ	
Detrended growth of foreign employment_ijt-1				-0.101* (0.058)
Measures of foreign market size				
Ln foreign employment_ijt-1	0.014 (0.01)		0.012** (0.006)	0.013 (0.009)
Ln foreign consumption_ijt-1		0.004** (0.002)		
Other control variables	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545
Number of Observations	272639	228830	227849	272639
Rho	-0.033	-0.009	-0.018	-0.033
	(0.032)	(0.007)	(0.018)	(0.031)
Log Likelihood	-3275.13	-2686.58	-2658.41	-3275.24

Table 7: Estimates of marginal effects for preliminary decisions: Country and industry fixed effects

Notes: Huber-White robust std errors in parentheses with \*\*\*,\*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Restrict col. (1) sample to	Restrict col. (1)	Add a multi-country
	antidumning filers	to large exporters	specif in col (1)
	(10)	(11)	(12)
	(10)	(11)	(12)
Stage 1: Industry's filing decision			
Measures of foreign demand			
Growth of foreign employment_ijt-1	-0.010***	-0.111***	-0.018***
	(0.003)	(0.034)	(0.006)
Measures of foreign market size			
Ln foreign employment_ijt-1	0.006***	0.025***	0.012***
	(0.000)	(0.005)	(0.001)
Other control variables	yes	yes	yes
Stage 2: Government's preliminary decision			
Measures of foreign demand			
Growth of foreign employment_ijt-1	-0.022	-0.361**	-0.057*
	(0.016)	(0.183)	(0.034)
Measures of foreign market size			
Ln foreign employment_ijt-1	0.001	0.032	0.005
	(0.002)	(0.021)	(0.004)
Other control variables	yes	yes	yes
Number of uncensored observations	545	379	545
Number of Observations	82905	62850	272639
Rho	-0.025**	0.034	-0.037**
	(0.012)	(0.196)	(0.018)
Log Likelihood	-3217.43	-2239.96	-3615.84

Table 8: Estimates of marginal effects for preliminary decisions: Additional specifications

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1.

Table 9: Predicted joint probability of a petition filing and a preliminary antidumping measure for a one s.d. fall in foreign demand

Spec.	Measure of foreign demand	Description	Predicted joint probability of a filing and a preliminary measure conditional on a one std. dev. fall in foreign demand
(1)	Employment growth	Basic specification	0.32%
(2)	Consumption growth	Basic specification	0.44%
(3)	Employment growth	Basic on spec . (2) sample	0.44%
(4)	Detrended employ. growth	Basic specification	0.29%
(5)	Detrended consump. growth	Basic specification	0.39%
(6)	Employment growth	Country & Industry F.E.s	0.51%
(7)	Consumption growth	Country & Industry F.E.s	0.23%
(8)	Employment growth	Country & Industry F.E.s	0.38%
(9)	Detrended employ. growth	Country & Industry F.E.s	0.49%
(10)	Employment growth	Restrict spec. (1) sample to industries with	0.64%
(11)	Employment growth	Restrict spec. (1) sample	0.92%
(12)	Employment growth	to large exporters Add a multi-country case dummy to spec. (1)	0.41%

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (14) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consumption growth as foreign demand shock
	(13)	(14)	(15)	(16)	(17)
Stage 1: Industry's filing decision Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.048*** (0.017)		-0.059*** (0.011)		
Growth of foreign consumption_ijt-1		-0.046** (0.019)			
Detrended growth of foreign employment_ijt-1				-0.045** (0.019)	0.020**
Moscures of foreign market size					(0.02)
Ln foreign employment_ijt-1	0.032*** (0.002)		0.024*** (0.002)	0.032*** (0.002)	
Ln foreign consumption_ijt-1		0.031*** (0.002)			0.031*** (0.002)
Other control variables	yes	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand					
Growth of foreign employment_ijt-1	0.017 (0.073)		0.033 (0.063)		
Growth of foreign consumption_ijt-1		0.036 (0.055)			
Detrended growth of foreign employment_it-1				-0.008 (0.077)	0.027
Measures of foreign market size					(0.058)
Ln foreign employment_ijt-1	0.016 (0.01)		0.007 (0.008)	0.016* (0.01)	
Ln foreign consumption_ijt-1		0.010 (0.012)			0.011 (0.012)
Other control variables	yes	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545	443
Number of Observations	272639	228830	227849	272639	228830
Rho	-0.095** (0.042)	-0.089* (0.048)	-0.084** (0.035)	-0.095** (0.042)	-0.090* (0.048)
Log Likelihood	-3678.159	-2987.146	-2973.175	-3678.728	-2988.404

#### Table 10: Estimates of marginal effects for final decisions

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Add country and industry fixed effects to col. (13)	Add country and industry fixed effects to col. (14)	Add country and industry fixed effects to col. (15)	Add country and industry fixed effects to col. (16)
	(18)	(19)	(20)	(21)
Stage 1: Industry's filing decision				
Measures of foreign demand				
Growth of foreign employment_ijt-1	-0.091**		-0.103***	
	(0.038)		(0.027)	
Growth of foreign consumption_ijt-1		-0.070**		
		(0.032)		
Detrended growth of foreign employment_ijt-1				-0.087**
Massuras of foreign market size				(0.039)
In foreign employment jit-1	0.068***		0 047***	0.066***
	(0.009)		(0.007)	(0.008)
Ln foreign consumption ijt-1	(0.000)	0.025***	(0.000)	()
о і <u>_</u> ,		(0.009)		
Other control variables	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand				
Growth of foreign employment_ijt-1	-0.073		-0.045	
	(0.164)		(0.138)	
Growth of foreign consumption_ijt-1		0.132		
Detrended growth of foreign employment, iit 1		(0.109)		0.009
Detrended growth of foreign employment_jt-1				-0.098
Measures of foreign market size				(0.130)
Ln foreign employment ijt-1	0.047		0.058*	0.052
	(0.034)		(0.031)	(0.033)
Ln foreign consumption_ijt-1		0.029		
		(0.036)		
Other control variables	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545
	5-5	UT.		5+5
Number of Observations	272639	228830	227849	272639
Rho	-0.059	-0.024	0.003	-0.041
	(0.101)	(0.106)	(0.088)	(0.096)
Log Likelihood	-3338.728	-2720.634	-2690.771	-3339.516

Table 11: Estimates of marginal effects for final decisions: Country and industry fixed effects

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Restrict col. (13) sample	Add a multi-country		
	to industries with	sample	case dummy to basic	
	antidumping filers	to large exporters	specif. in col. (13)	
	(22)	(23)	(24)	
Stage 1: Industry's filing decision				
Measures of foreign demand				
Growth of foreign employment_ijt-1	-0.045***	-0.023***	-0.051***	
	(0.015)	(0.007)	(0.018)	
Measures of foreign market size				
Ln foreign employment_ijt-1	0.028***	0.005***	0.034***	
	(0.002)	(0.001)	(0.002)	
Other control variables	yes	yes	yes	
Stage 2: Government's preliminary decision				
Measures of foreign demand				
Growth of foreign employment_ijt-1	0.021	0.017	0.017	
	(0.059)	(0.026)	(0.079)	
Measures of foreign market size				
Ln foreign employment_ijt-1	0.008	0.004	0.019*	
	(0.009)	(0.004)	(0.011)	
Other control variables	yes	yes	yes	
Number of uncensored observations	545	379	545	
Number of Observations	82905	62850	272639	
Rho	-0.103**	-0.061*	-0.101**	
-	(0.048)	(0.033)	(0.045)	
Log Likelihood	-3279.807	-2290.329	-3676.879	

Table 12: Estimates of marginal effects for final decisions: Additional specifications

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1.

Spec.	Measure of foreign demand	Description	Predicted joint probability of a filing and a final measure conditional on a one std. dev. fall in foreign demand
(13)	Employment growth	Basic specification	0.52%
(14)	Consumption growth	Basic specification	0.69%
(15)	Employment growth	Basic on spec . (14) sample	0.56%
(16)	Detrended employ. growth	Basic specification	0.48%
(17)	Detrended consump. growth	Basic specification	0.49%
(18)	Employment growth	Country & Industry F.E.s	0.90%
(19)	Consumption growth	Country & Industry F.E.s	1.00%
(20)	Employment growth	Country & Industry F.E.s	0.91%
(21)	Detrended employ. growth	Country & Industry F.E.s	0.84%
(22)	Employment growth	Restrict spec. (13) sample to industries with	0.71%
(23)	Employment growth	Restrict spec. (13) sample	0.45%
(24)	Employment growth	Add a multi-country case dummy to spec. (13)	0.55%

Table 13: Predicted joint probability of a petition filing and a final antidumping measure for a one s.d. fall in foreign demand

## References

- Bown, Chad P. and Meredith A. Crowley. 2007. "Trade Deflection and Trade Depression," Journal of International Economics 72(1): 176-201.
- [2] Baldwin, Robert and Steagall, Jeffrey W. 1994. "An Analysis of ITC Decisions in Antidumping, Countervailing Duty and Safeguard Cases." Weltwirtschaftliches Archiv, 130:290-307.
- Blonigen, Bruce A. 2006. "Evolving Discretionary Practices of US Antidumping Activity." Canadian Journal of Economics, 39: 874-900.
- [4] Blonigen, Bruce A. and Bown, Chad P. 2003. "Antidumping and Retaliation Threats." Journal of International Economics, 60: 249-273.
- [5] Blonigen, Bruce A. and Haynes, Stephen E. 2002. "Antidumping Investigations and the Pass-Through of Antidumping Duties and Exchange Rates." *American Economic Review*, 92: 1044-1061.
- [6] Blonigen, Bruce A. and Park, Jee-Hyeong 2004. "Dynamic Pricing in the Presence of Antidumping Policy: Theory and Evidence." American Economic Review, 94: 134-154.
- Bown, Chad August 2007. "China's WTO Entry: Antidumping, Safeguards, and Dispute Settlement." NBER Working Paper # 13349.
- [8] Bown, Chad and Crowley, Meredith. June 2007. "China's Export Growth and the China Safeguard: Threats to the World Trading System?" Federal Reserve Bank of Chicago Working Paper # 2004-28.
- [9] Cadot, Olivier; de Melo, Jaime; and Tumurchudur, Bolormaa 2007. "Anti-dumping Sunset Reviews: The Uneven Reach of WTO Disciplines." unpublished mimeo.
- [10] Crowley, Meredith A. 2007. "Split Decisions in Antidumping Cases." Federal Reserve Bank of Chicago mimeo.

- [11] Ethier, Wilfred J. 1982. "Dumping." Journal of Political Economy, 90:487-506.
- Feinberg, Robert M. 2005. "U.S. Antidumping Enforcement and Macroeconomic Indicators Revisited: Do Petitioners Learn?" *Review of World Economics*, 141: 612-622.
- [13] Feinberg, Robert M. and Reynolds, Kara M. 2006. "The Spread of Antidumping Regimes and the Role of Retaliation in Filings." Southern Economic Journal.
- [14] Hansen, Wendy L. 1990. "The International Trade Commission and the Politics of Protectionism." American Political Science Review, 84: 21-46.
- [15] Hansen, Wendy L. and Prusa, Thomas J. 1996. "Cumulation and ITC Decision Making: The Sum of the Parts is Greater than the Whole." *Economic Inquiry*, 34: 746-69.
- [16] Hansen, Wendy L. and Prusa, Thomas J. 1997. "The Economics and Politics of Trade Policy: An Empirical Analysis of ITC Decision Making." *Review of International Economics*, 5: 230-45.
- [17] Hartigan, James C. 1996. "Predatory dumping." Canadian Journal of Economics, 29: 228-239.
- [18] Jallab, Mustapha Sadni; Sandretto, Rene; and Gbakou, Monnet Benoit Patrick 2006. "Antdiumping Procedures and Macroeconomic Factors: A Comparison between the United States and the European Union." *Global Economy Journal*, 6: Article 5.
- [19] Knetter, Michael M. and Prusa, Thomas J. 2003. "Macroeconomic Factors and Antidumping Filings: Evidence from Four Countries." *Journal of International Economics*, 61:1-17.
- [20] Miranda, Jorge; Torres, Raul A.; and Ruiz, Mario. 1998. "The international use of antidumping: 1987-1999." Journal of World Trade, 32: 5-71.
- [21] Moore, Michael O. 1992. "Rules or Politics? An Empirical Analysis of ITC Anti-dumping Decisions." *Economic Inquiry*, 30: 449-66.

- [22] Prusa, Thomas J. 1997. "The Trade Effects of U.S. Antidumping Actions," in *The Effects of U.S. Trade Protection and Promotion Policies* Robert C. Feenstra, ed. Chicago: University of Chicago Press.
- [23] Prusa, Thomas J. and Skeath, Susan 2002. "The economic and strategic motives for antidumping filings." *Review of World Economics*, 138: 389-413.
- [24] Staiger, Robert W. and Wolak, Frank A. 1992. "The effect of domestic antidumping law in the presence of foreign monopoly." *Journal of International Economics*, 32:265-287.
- [25] Staiger, Robert W. and Wolak, Frank A. 1994. "Measuring Industry-Specific Protection: Antidumping in the United States." Brookings Papers on Economic Activity: Microeconomics, 51-118.
- [26] Van de Ven, P.M.M. and Van Praag, Bernard M.S. 1981. "The Demand for Deductibles in Private Health Insurance: A Probit Model with Sample Selection." *Journal of Econometrics*, 17:229-252.

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (26) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consumption growth as foreign demand shock
	(25)	(26)	(27)	(28)	(29)
<u>Stage 1: Industry's filing decision</u> Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.016*** (0.006)		-0.028*** (0.005)		
Growth of foreign consumption_ijt-1		-0.017** (0.007)			
Detrended growth of foreign employment_ijt-1				-0.013** (0.006)	
Detrended growth of foreign consumption_ijt-1					-0.014** (0.007)
Measures of foreign market size	0.011***		0.011***	0.009***	
	(0.001)		(0.001)	(0.001)	
In foreign consumption_ijt-1		0.011*** (0.001)			0.011*** (0.001)
Other control variables	yes	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.055* (0.03)		-0.050 (0.037)		
Growth of foreign consumption_ijt-1		-0.002 (0.019)			
Detrended growth of foreign employment_ijt-1				-0.063** (0.029)	
Detrended growth of foreign consumption_ijt-1					-0.004 (0.019)
Measures of foreign market size Ln foreign employment ijt-1	0.005		0.005	0.004	
	(0.003)		(0.004)	(0.003)	
Ln foreign consumption_ijt-1		0.003 (0.004)			0.003 (0.004)
Other control variables	yes	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545	443
Number of Observations	272639	228830	227849	272639	228830
Rho	-0.029* (0.016)	-0.031* (0.018)	-0.032* (0.018)	-0.028** (0.014)	-0.031* (0.019)
Log Likelihood	-3619.513	-2954.253	-2939.382	-3619.192	-2955.446

## Appendix Table A: Estimates of marginal effects for a preliminary antidumping duty

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (31) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consumption growth as foreign demand shock
	(30)	(31)	(32)	(33)	(34)
<u>Stage 1: Industry's filing decision</u> Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.047*** (0.016)		-0.068*** (0.013)		
Growth of foreign consumption_ijt-1		-0.052** (0.021)			
Detrended growth of foreign employment_ijt-1				-0.044** (0.019)	
Detrended growth of foreign consumption_ijt-1					-0.044** (0.022)
Measures of foreign market size					
Ln foreign employment_ijt-1	0.031***		0.028***	0.031***	
Ln foreign consumption_ijt-1	(0.002)	0.035*** (0.003)	(0.002)	(0.002)	0.035*** (0.003)
Other control variables	yes	yes	yes	yes	yes
Stage 2: Government's preliminary decision Measures of foreign demand					
Growth of foreign employment_ijt-1	0.018 (0.072)		0.038 (0.074)		
Growth of foreign consumption_ijt-1		0.042 (0.062)			
Detrended growth of foreign employment_ijt-1				-0.003 (0.076)	
Detrended growth of foreign consumption_ijt-1					0.037 (0.064)
Measures of foreign market size Ln foreign employment_ijt-1	0.014		0.007	0.014	
	(0.01)		(0.009)	(0.01)	
Ln foreign consumption_ijt-1		0.011 (0.013)			0.011 (0.013)
Other control variables	yes	yes	yes	yes	yes
Number of uncensored observations	545	443	441	545	443
Number of Observations	272639	228830	227849	272639	228830
Rho	-0.093**	-0.090*	-0.087**	-0.093**	-0.090*
	(0.041)	(0.053)	(0.041)	(0.042)	(0.053)
Log Likelihood	-3679.57	-2988.10	-2974.43	-3680.16	-2989.33

### Appendix Table B: Estimates of marginal effects for a final antidumping duty

Notes: Huber-White robust std errors in parentheses with \*\*\*, \*\*, and \* indicating statistical significance at the 1%, 5% and 10% levels. Other control variables in the stage 2: government decision include: the growth of import penetration\_ijt-1, the growth of domestic employment\_it-1, capacity utilization\_it-1, the growth of US GDP\_t-1, and the bilateral real exchange rate between the dollar and country j's currency\_t-1. Other control variables in the stage 1: Industry filing decision include all controls in the stage 2 decison plus import penetration\_ijt-1, the logged level of domestic employment\_it-1, and value added/output\_it-1.

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (2) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consump. growth as foreign demand shock
	(1)	(2)	(3)	(4)	(5)
Stage 1: Industry's filing decision					
Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.211***		-0.337***		
	(0.073)		(0.066)		
Growth of foreign consumption_ijt-1		-0.208**			
		(0.086)			
Detrended growth of foreign emp_ijt-1				-0.195**	
				(0.083)	
Detrended growth of foreign cons_ijt-1					-0.176**
					(0.088)
Measures of foreign market size					
Ln foreign employment_ijt-1	0.139***		0.140***	0.138***	
	(0.009)		(0.010)	(0.009)	
Ln foreign consumption_ijt-1		0.139***			0.139***
		(0.010)			(0.010)
Industry-level control variables					
Growth of country j's import pen_ijt-1	0.026***	0.033***	0.028**	0.026***	0.033***
	(0.009)	(0.011)	(0.011)	(0.009)	(0.011)
Import penetration_ijt-1	2.576***	2.786***	2.675***	2.574***	2.783***
	(0.154)	(0.181)	(0.180)	(0.154)	(0.181)
Growth of domestic employment_it-1	0.165	0.273	0.324*	0.162	0.269
	(0.164)	(0.184)	(0.178)	(0.164)	(0.184)
Ln(employment)_ijt-1	0.263***	0.257***	0.262***	0.263***	0.257***
	(0.014)	(0.015)	(0.015)	(0.014)	(0.015)
Capacity utilization_it-1	-0.343***	-0.356***	-0.369***	-0.344***	-0.357***
	(0.034)	(0.038)	(0.038)	(0.034)	(0.038)
Value added/output_it-1	-1.593***	-1.493***	-1.608***	-1.599***	-1.501***
	(0.136)	(0.145)	(0.153)	(0.135)	(0.144)
Macroeconomic control variables					
Growth of US GDP_t-1	2.192***	0.649	1.116	2.195***	0.663
	(0.824)	(0.954)	(0.912)	(0.824)	(0.956)
Ln (bilateral real exchange rate)_jt-1	0.032***	0.042***	0.034***	0.032***	0.041***
	(0.005)	(0.006)	(0.006)	(0.005)	(0.006)
Constant	-4.012***	-4.607***	-3.950***	-4.000***	-4.601***
	(0.147)	(0.199)	(0.161)	(0.147)	(0.199)
Number of Observations	272639	228830	227849	272639	228830
	2,2000		0	2,2000	
Rho	-0.435**	-0.417*	-0.432*	-0.457**	-0.415*
-	(0.213)	(0.231)	(0.224)	(0.210)	(0.231)
	(0.220)	(	(	(	(
Log Likelihood	-3616.14	-2952.25	-2937.34	-3616.09	-2953.45
AIC	7272.283	5944.499	5914.671	7272.180	5946.892
BIC	7482.601	6151.313	6121.400	7482.499	6153.706

## Appendix Table C: ML coefficient estimates for table 6: Stage 1

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (2) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consump. growth as foreign demand shock
	(1)	(2)	(3)	(4)	(5)
Stage 2: Government's preliminary decision					
Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.649* (0.389)		-0.589 (0.452)		
Growth of foreign consumption_ijt-1		-0.005 (0.227)			
Detrended growth of foreign emp_ijt-1				-0.807* (0.419)	
Detrended growth of foreign cons_ijt-1					-0.038 (0.231)
Measures of foreign market size					
Ln foreign employment_ijt-1	0.057 (0.045)		0.056 (0.049)	0.054 (0.044)	
Ln foreign consumption_ijt-1		0.028 (0.054)			0.029 (0.054)
Industry-level control variables					
Growth of country j's import pen_ijt-1	-0.001 (0.079)	0.034 (0.095)	0.032 (0.088)	-0.003 (0.079)	0.034 (0.095)
Growth of domestic employment_it-1	-1.263 (0.795)	-0.781 (0.908)	-0.684 (0.897)	-1.254 (0.791)	-0.789 (0.910)
Capacity utilization_it-1	0.263*** (0.097)	0.283*** (0.109)	0.272** (0.110)	0.265*** (0.096)	0.282** (0.110)
Macroeconomic control variables					
Growth of US GDP_t-1	2.346 (2.812)	1.481 (3.160)	1.982 (3.210)	2.346 (2.784)	1.479 (3.162)
Ln (bilateral real exchange rate)_jt-1	-0.006 (0.023)	-0.002 (0.025)	0.001 (0.024)	-0.008 (0.023)	-0.002 (0.025)
Constant	0.557	0.654	0.479	0.644	0.642
Number of Uncensored Obs.	545	443	441	545	443

## Appendix Table D: ML coefficient estimates for table 6: Stage 2

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (14) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consump. growth as foreign demand shock
	(13)	(14)	(15)	(16)	(17)
Stage 1: Industry's filing decision Measures of foreign demand					
Growth of foreign employment_ijt-1	-0.211*** (0.074)		-0.338*** (0.066)		
Growth of foreign consumption_ijt-1	(0.07.1)	-0.208** (0.086)	(0.000)		
Detrended growth of foreign emp_ijt-1		()		-0.195** (0.083)	
Detrended growth of foreign cons_ijt-1				()	-0.176** (0.088)
Measures of foreign market size					. ,
Ln foreign employment_ijt-1	0.139***		0.140***	0.138***	
	(0.009)	0. 4 0.0 th th th	(0.010)	(0.009)	0. 4 0.0 t t t t
In foreign consumption_ijt-1		0.139***			0.139***
Industry-level control variables		(0.010)			(0.010)
Growth of country i's import pen jit-1	0.026***	0.033***	0.028**	0.026***	0.033***
	(0.009)	(0.011)	(0.011)	(0.009)	(0.011)
Import penetration_ijt-1	2.564***	2.762***	2.652***	2.562***	2.759***
	(0.153)	(0.180)	(0.178)	(0.153)	(0.180)
Growth of domestic employment_it-1	0.167	0.275	0.327*	0.163	0.271
	(0.164)	(0.184)	(0.178)	(0.164)	(0.184)
Ln(employment)_ijt-1	0.262***	0.256***	0.261***	0.262***	0.256***
	(0.014)	(0.015)	(0.015)	(0.014)	(0.015)
Capacity utilization_it-1	-0.343***	-0.356***	-0.369***	-0.344***	-0.35/***
Value added/output_it_1	(0.034)	(U.U37) _1 517***	(0.038)	(0.034)	(U.U38) _1 525***
value added/output_it=1	(0 134)	(0 144)	(0 151)	(0 134)	(0 143)
Macroeconomic control variables	(0.131)	(0.111)	(0.131)	(0.13 1)	(0.1.13)
Growth of US GDP_t-1	2.194***	0.655	1.121	2.196***	0.669
_	(0.824)	(0.953)	(0.911)	(0.824)	(0.955)
Ln (bilateral real exchange rate)_jt-1	0.032***	0.042***	0.034***	0.032***	0.041***
	(0.005)	(0.006)	(0.006)	(0.005)	(0.006)
Constant	-4.004***	-4.594***	-3.935***	-3.992***	-4.589***
	(0.148)	(0.20)	(0.162)	(0.147)	(0.20)
Number of Observations	272639	228830	227849	272639	228830
Rho	-0.419**	-0.403*	-0.480**	-0.417**	-0.400*
	(0.186)	(0.215)	(0.202)	(0.184)	(0.214)
Log Likelihood	-3678.159	-2987.146	-2973.175	-3678.728	-2988.404
AIC	7396.319	6014.293	5986.35	7397.455	6016.807
BIC	7606.637	6221.107	6193.079	7607.773	6223.622

## Appendix Table E: ML coefficient estimates for table 10: Stage 1

	Basic specific.	Substitute foreign consump. growth	Basic spec. on col. (14) sample	Detrended foreign employment growth as foreign demand shock	Detrended foreign consump. growth as foreign demand shock
	(13)	(14)	(15)	(16)	(17)
Stage 2: Government's preliminary decision Measures of foreign demand					
Growth of foreign employment_ijt-1	0.077		0.189		
	(0.322)		(0.362)		
Growth of foreign consumption_ijt-1		0.160 (0.250)			
Detrended growth of foreign emp_ijt-1				-0.033 (0.336)	
Detrended growth of foreign cons_ijt-1					0.121 (0.257)
Measures of foreign market size					
Ln foreign employment_ijt-1	0.069		0.041	0.071*	
	(0.044)		(0.047)	(0.043)	
Ln foreign consumption_ijt-1		0.047 (0.052)			0.048 (0.052)
Industry-level control variables					
Growth of country j's import pen_ijt-1	-0.170*	-0.162*	-0.176**	-0.171*	-0.162*
	(0.088)	(0.087)	(0.088)	(0.088)	(0.087)
Growth of domestic employment_it-1	0.277	0.285	0.338	0.292	0.295
	(0.753)	(0.887)	(0.847)	(0.753)	(0.888)
Capacity utilization_it-1	0.123	0.170*	0.183**	0.122	0.170*
	(0.077)	(0.090)	(0.085)	(0.076)	(0.090)
Macroeconomic control variables					
Growth of US GDP_t-1	2.316	3.840	3.924	2.382	3.850
	(2.623)	(3.023)	(2.992)	(2.623)	(3.027)
Ln (bilateral real exchange rate)_jt-1	0.026	0.035	0.034	0.027	0.036
	(0.021)	(0.023)	(0.023)	(0.021)	(0.023)
Constant	-0 113	-0 25/	0 102	-0 1/12	-0 274
Constant	(0.201)	(1 202)	(0.845)	(0 791)	(1 203)
Number of Uncensored Obs.	545	443	441	545	443

## Appendix Table F: ML coefficient estimates for table 10: Stage 2